Author's response to reviews

Title: The Potential Oral Health Impact of Cost Barriers to Dental Care: Findings from a Canadian Population-Based Study

Authors:

Brandy Thompson (brandy.thompson@mail.utoronto.ca)
Peter Cooney (peter.cooney@hc-sc.gc.ca)
Herenia Lawrence (herenia.lawrence@dentistry.utoronto.ca)
Vahid Ravaghi (vahid.ravaghi@mcgill.ca)
Carlos Quiñonez (carlos.quinonez@utoronto.ca)

Version: 2
Date: 24 January 2014

Author's response to reviews: see over
January 23, 2014

To: Journal Editorial Office, BioMed Central

Re: Revised Manuscript Submission - “The Potential Oral Health Impact of Cost Barriers to Dental Care: Findings from a Canadian Population-Based Study”

Dear Ms. Eloisa Nolasco,

Please accept for review our revised manuscript entitled: The Potential Oral Health Impact of Cost Barriers to Dental Care: Findings from a Canadian Population-Based Study (Manuscript 1358668225115212).

As per your feedback, changes have been made to the original submission. Details pertaining to ethics approval and data management have been added to the “methods” section. Duplicate data including the first two tables in the “results” section have been removed and reworded. Reference has been made to the other recently published manuscript (Thompson et al, 2014). The “limitations” section has also been reworded to avoid duplication.

Please do not hesitate to contact me should you have any questions or further feedback.

We thank you for your time and look forward to your response,

Brandy Thompson (Corresponding Author)
brandy.thompson@mail.utoronto.ca
905-466-8813